MA194 { Using Unix

Instructor: Dan Chaney (chaney@ms.uky.edu) Office hours: Mon 2-3, 606-257-2150

Class #17 - Tuesday, March 19, 1996 – Week 9

In-class hotline: (606) 323–3681 Community Colleges: Tues 9:00-9:50 & 10:00-10:50 PM EST UK/Lexington: Wed & Fri 9:00 PM EST, TCI Channel 16

URL: http://www.ms.uky.edu/~ma194

Mailing list: ma194-class@ms.uky.edu To join the mailing list, send a note to ma194@ms.uky.edu

Tonight’s Topics ffl Mail problems ffl File systems (Permissions) ffl Shell scripts